\documentclass{svjour3}
\usepackage{cite}

\begin{document}

I must not fear. Fear is the mind-killer. Fear is the little-death that brings total obliteration. I will face my fear. I will permit it to pass over me and through me. And when it has gone past I will turn the inner eye to see its path. Where the fear has gone there will be nothing. Only I will remain.
\cite{KORU2005}
\cite{SUNDARAM2005}
\cite{BOETTICHER2005}
\cite{MAIR2005}
\cite{MENZIES2005}
\cite{CHEN2005}
\cite{MENZIES2005-2}
\cite{HAYES2005}
\cite{HAYES2005-2}
\cite{JALALI2007}
\cite{MA2007}
\cite{DEKHTYAR2007}
\cite{LIU2007}
\cite{WEISS07}
\cite{MIZUNO2007}
\cite{GOKHALE2008}
\cite{AZZEH2008}
\cite{KORTE2008}
\cite{MENZIES2008}
\cite{JIANG2008}
\cite{MOSER2008}
\cite{BOETTICHER08}
\cite{JALALI08}
\cite{LI08}
\cite{LIEBCHEN08}
\cite{WATANABE08}
\cite{BRAGA08}
\cite{PORT2008}
\cite{JIANG20082}
\cite{WEDEL2008}
\cite{TOSUN2008}
\cite{KULTUR2008}
\cite{HATA08}
\cite{KAMEI08}
\cite{YANG08}
\cite{RANA2008}
\cite{HECKMAN2008}
\cite{BRODER2008}
\cite{HOLZ2008}
\cite{ERDOGMUS2008}
\cite{BOETTICHER2009}
\cite{KITCHENHAM2009}
\cite{GAY2009}
\cite{TOSUN2009}
\cite{TOSUN2009-3}
\cite{JIANG2009}
\cite{AZZEH2009}
\cite{MENDE2009}
\cite{WAGNER2009}
\cite{FU09}
\cite{MENZIES2009}
\cite{SARCIA2009-2}
\cite{MORASCA2009}
\cite{SARCIA2009}
\cite{BIRD2009}
\cite{CAGLAYAN2009}
\cite{TOSUN20092}
\cite{WINKLER2009}
\cite{SINGH2009}
\cite{CATAL2009}
\cite{MORAN2009}
\cite{JIA2009}
\cite{SAMI2010}
\cite{MALHOTRA2010}
\cite{FERNANDES2010}

\bibliographystyle{ieeetr}
\bibliography{references}

\end{document}